1968



JGR Biogeosciences

RESEARCH ARTICLE

10.1029/2022JG007210

Key Points:

- Dinitrogen dominated nitrogen gas emissions from tropical forest soils
 Valleys were hot spots of nitrogen
- emissions in this tropical landscape
 Dinitrogen fluxes may change our
 understanding of nitrogen budgets it
- understanding of nitrogen budgets in this forest

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

M. Almaraz, almaraz@nceas.ucsb.edu

Citation:

Almaraz, M., Groffman, P. M., Silver, W. L., Hall, S. J., Lin, Y., O'Connell, C., & Porder, S. (2023). Dinitrogen emissions dominate nitrogen gas emissions from soils with low oxygen availability in a moist tropical forest. *Journal of Geophysical Research: Biogeosciences*, *128*, e2022/G007210. https://doi. org/10.1029/2022JG007210

Received 27 SEP 2022 Accepted 15 DEC 2022

Author Contributions:

Conceptualization: Maya Almaraz, Peter M. Groffman, Stephen Porder Data curation: Maya Almaraz Formal analysis: Yang Lin Funding acquisition: Maya Almaraz, Stephen Porder Investigation: Maya Almaraz, Whendee L. Silver, Steven J. Hall Methodology: Maya Almaraz, Peter M. Groffman Resources: Peter M. Groffman, Whendee L. Silver, Steven J. Hall, Christine O'Connell Software: Yang Lin Supervision: Peter M. Groffman, Stephen Porder Validation: Christine O'Connell Visualization: Maya Almaraz Writing - original draft: Maya Almaraz Writing - review & editing: Peter M. Groffman, Whendee L. Silver, Steven J. Hall, Yang Lin, Christine O'Connell, Stephen Porder

© 2023. American Geophysical Union. All Rights Reserved.

Dinitrogen Emissions Dominate Nitrogen Gas Emissions From Soils With Low Oxygen Availability in a Moist Tropical Forest

Maya Almaraz^{1,2} ^(D), Peter M. Groffman^{2,3}, Whendee L. Silver⁴ ^(D), Steven J. Hall⁵ ^(D), Yang Lin^{4,6} ^(D), Christine O'Connell⁷ ^(D), and Stephen Porder⁸

¹High Meadows Environmental Institute, Princeton University, Princeton, NJ, USA, ²Cary Institute of Ecosystem Studies, Millbrook, NY, USA, ³Advanced Science Research Center at the Graduate Center, City University of New York, New York, NY, USA, ⁴Department of Environmental Science, Policy and Management, University of California, Berkeley, CA, USA, ⁵Department of Ecology, Evolution and Organismal Biology, Iowa State University, Ames, IA, USA, ⁶Department of Soil and Water Sciences, University of Florida, Gainesville, FL, USA, ⁷Department of International Development, Macalester University, Saint Paul, MN, USA, ⁸Department of Ecology and Evolutionary Biology, Brown University, Providence, RI, USA

Abstract Lowland tropical forest soils are relatively N rich and are the largest global source of N_2O (a powerful greenhouse gas) to the atmosphere. Despite the importance of tropical N cycling, there have been few direct measurements of N_2 (an inert gas that can serve as an alternate fate for N_2O) in tropical soils, limiting our ability to characterize N budgets, manage soils to reduce N_2O production, or predict the future role that N limitation to primary productivity will play in buffering against climate change. We collected soils from across macro- and micro-topographic gradients that have previously been shown to differ in O_2 availability and trace gas emissions. We then incubated these soils under oxic and anoxic headspaces to explore the relative effect of soil location versus transient redox conditions. No matter where the soils came from, or what headspace O_2 was used in the incubation, N_2 emissions dominated the flux of N gas losses. In the macrotopography plots, production of N_2 and N_2O were higher in low O_2 valleys than on more aerated ridges and slopes. In the microtopography plots, N_2 emissions from plots with lower mean soil O_2 (5%–10%) were greater than in plots with higher mean soil O_2 (10%–20%). We estimate an N gas flux of ~37 kg N/ha/yr from this forest, 99% as N_2 . These results suggest that N_2 fluxes may have been systematically underestimated in these landscapes, and that the measurements we present call for a reevaluation of the N budgets in lowland tropical forest ecosystems.

Plain Language Summary Nitrogen is a macronutrient that limits plant productivity in much of the world. Tropical forests have a relatively large number of nitrogen fixing species, they tend to cycle nitrogen in excess of plant demand. This excess nitrogen is released into the environment in either dissolved or gaseous forms. Wet, warm, and organic matter-rich soils of tropical forests result in a hot spot for denitrification; a process that removes nitrogen from soils and converts in into either nitric oxide, nitrous oxide, or dinitrogen gas. Because dinitrogen makes up 80% of earth's atmosphere, it's difficult to measure a small flux coming off the soil. This knowledge gap has resulted in a limited understanding of soil nitrogen cycling in tropical ecosystems. Because soil moisture can influence denitrification, we tested how wet points in the landscape influence emissions of dinitrogen gas. We found that wetter low points in the landscape result in the highest dinitrogen fluxes, and that dinitrogen dominated nitrogen gas emissions from these soils. We used these data to estimate dinitrogen emissions for this forest and generated a larger value than previously calculated. Such results suggest a need to integrate dinitrogen measurements into studies of nitrogen cycling in tropical forest soils.

1. Introduction

Nitrogen (N) is the nutrient that most frequently limits primary productivity. The nutrient status of tropical forest soils are a key determinate in whether they will continue to act as a carbon (C) sink in the face of rising temperatures and atmospheric CO₂ concentrations (Oren et al., 2001; Y. P. Wang & Houlton, 2009). Lowland tropical soils are also an important source of greenhouse emissions to the atmosphere. As a result of circulating large amounts of N annually, tropical soils experience high emission rates in the form of nitric oxide (NO), nitrous oxide (N₂O—a powerful greenhouse gas i.e., responsible for ~10% of current anthropogenic warming), and dinitrogen (N₂—an inert and environmentally benign gas; Cleveland et al., 1999; Firestone & Davidson, 1989;



Vitousek & Matson, 1988; Vitousek & Sanford, 1986; Templer et al., 2008). As such, tropical forest soils are the largest source of N_2O to the atmosphere (Zhuang et al., 2012). Determining the current and future N status of soils requires an understanding of how inputs balance with outputs to produce net accumulation or loss. A major impediment to improving this understanding is our ability to measure N_2 emissions from soils. Dinitrogen is not only the largest pool of N on earth, but it's flux from soils is incredibly difficult to measure given its high atmospheric background. Despite the importance of N_2 to understanding soil nutrient status and its influence on the future climate, few direct measurements of N_2 have been made from tropical forest soils (Almaraz et al., 2020), limiting our ability to understand the spatial distribution and magnitude of this flux.

Dinitrogen is typically produced through the process of microbial denitrification, which converts soil nitrate (NO_3^{-}) to N_2 under anaerobic conditions. This process requires C to act as an electron donor, soil NO_3^{-} as a substrate, and redox conditions that allow for complete denitrification, with the latter typically driven by low soil O_2 conditions, high soil moisture, or soil aggregates that result in anaerobic microsites. In tropical soils where soil O_2 is depleted, soil redox conditions may favor the reduction of N_2O to N_2 before the gas escapes to the atmosphere. However, without adequate redox conditions, incomplete denitrification can result in the production of either NO or, more commonly in moist tropical forest soils, N_2O . Thus, both N_2O and N_2 can be produced during anaerobic denitrification (Stein, 2019). Nitrous oxide can also be produced during aerobic nitrification as well as during other processes, such as anaerobic dissimilatory nitrate reduction to ammonium (Butterbach-Bahl et al., 2013; Silver et al., 2001) and iron-coupled ammonium oxidation (Feammox; Yang et al., 2012). The relative amounts of N_2O and N_2 emitted from soils are thus also thought to be influenced by controls on these processes such as soil redox potential, N substrate, and C availability (Davidson et al., 2000; Groffman & Tiedje, 1989; Weier et al., 1993).

While redox is undoubtedly an important driver of the N₂O yield from tropical forests soils, testing this control in the field has been difficult (Butterbach-Bahl et al., 2013; Firestone & Davidson, 1989; Firestone et al., 1980; Groffman & Tiedje, 1989; Weier et al., 1993). Given the high spatial and temporal variability of soil redox in natural ecosystems, it is perhaps not surprising that our understanding of the ratio of N₂O:N₂ emissions is poorly constrained. This gap in understanding is particularly relevant in tropical forests, where N gas losses are thought to be high (Houlton et al., 2006), but where factors that influence soil N gas emissions may also be highly variable (Townsend et al., 2008). Quantifying N gas fluxes from tropical forests is a key step in understanding N cycling and availability in tropical forests.

A recent review of denitrification studies found an overreliance on laboratory manipulation studies (e.g., studies that add C, N, or water to soils) and suggested that measurements made across natural gradients could help to overcome knowledge gaps (Almaraz et al., 2020). While many studies have measured N gas emissions in response to laboratory manipulated O_2 (Almaraz et al., 2020), few have observed how N gas emissions vary across natural soil redox gradients. Because O_2 diffuses 10,000 times more slowly through water than through air, soil moisture is a primary control on redox (Ponnamperuma, 1972). Soil water content is largely a function of rainfall, evapotranspiration, soil texture and aggregation. Soils microsites that remain saturated for longer periods are more likely to support complete denitrification to N_2 , rather than having N_2O escape to the atmosphere prior to complete reduction (Bremner & Blackmer, 1981; Goreau et al., 1980; Poth & Focht, 1985). However, even in well-aerated soils, denitrification can occur in anaerobic microsites where biological activity depletes O_2 supply. Anaerobic microsites in well aerated soils are more common in well-aggregated or clay rich soils and those with high rates of decomposition (Parkin, 1987). The prevalence of such textural and microsite characteristics varies somewhat predictably with topography, as a result of erosion, deposition, illuviation and eluviation.

Topographic gradients provide an excellent way to explore the link between N gas emissions and soil redox. In the Luquillo Experimental Forest in Puerto Rico, soil O_2 varies predictably with topographic position and affects rates of N cycling and loss (Hall et al., 2013; O'Connell et al., 2018; Pett-Ridge et al., 2006; Silver et al., 1999, 2013). The Luquillo landscape is characterized by highly dissected topography, where "knife-edge" ridge tops drain into steep gradient ephemeral streams and form small first- and zero-order valleys (Scatena, 1989). Extreme precipitation events in these riparian valleys can cause preferential removal of clay-size particles, increasing sand content in valleys and causing topographic texture patterns to be reversed from that of typically clay rich valleys (Hall et al., 2015; Hook & Burke, 2000; O'Connell *personal communication*). Despite high annual precipitation, high soil drainage rates result in soils that are rarely inundated, making saturated microsites within aggregates particularly important for N gas production (Almaraz et al., 2019; Hall et al., 2013). High biological activity also plays an important role in these soils, as O_2 consumption can often exceed diffusive resupply (Silver et al., 1999).





Figure 1. (a) The Luquillo Experimental Forest, located in northeastern Puerto Rico. (b) Macrotopography (ridge, slope, valley) plots near the El Verde field station in the Sonadora watershed, and (c) microtopography (slope plots that differ in soil oxygen availability) plots located in the Bisley 3 Watershed. Adapted from Harris et al. (2012).

Methodological challenges associated with measuring soil N_2 flux into an atmosphere that is 78% N_2 make it difficult to partition N gas emissions between their various forms. Many N_2 measurement techniques involve the use of inhibitors that have ancillary effects on controlling processes, alteration of substrate levels, or disruption of soil physical conditions (Almaraz et al., 2020; Groffman et al., 2006). Because N gas emissions are dependent on soil aggregation, anaerobic microsites, biological activity, and moisture and drainage, understanding the relative emissions of different N gas species will require measuring them with methods that maintain conditions that are as similar as possible to those in the field.

In this study, we used a laboratory-based gas flow incubation system that allows direct measurement of N_2 and N_2O emissions from intact soil cores under controlled headspace O_2 conditions in the laboratory. We asked whether N gas emissions vary predictably from soils collected in different topographic positions and whether this variation also occurs with short-term manipulations of headspace O_2 during laboratory incubations. We analyzed all soils under a range of headspace O_2 concentrations (0%, 5%, 10%, 20%), and hypothesized that, (a) the N_2 :N₂O ratio would be inversely related to headspace O_2 concentration, (b) valleys and low points in the landscape (i.e., low soil O_2 plots) would be hot-spots for N gas production, and (c) both N_2 and N_2O emissions would be negatively correlated with soil O_2 availability, which co-varies with soil moisture, topography and soil texture in this landscape (Silver et al., 1999, 2013).

2. Methods

2.1. Study Location

This study was conducted in the Luquillo Experimental Forest (LEF), a U.S. National Science Foundation funded Critical Zone Observatory and Long Term Ecological Research site in the USDA Forest Service managed El Yunque National Forest in northeastern Puerto Rico (Figure 1a). The LEF is a premontane tropical forest land-scape that receives 3–4 m of rain annually with a mean annual temperature of 23°C (Garcia-Montino et al., 1996; Scatena, 1989). Rainfall at the time of sampling was typical of this location and preceded the 2015 drought, which began in May of 2015 (O'Connell et al., 2018). We sampled in forests dominated by Tabonuco (*Dacryodes excelsa* Vahl) and underlain by volcaniclastic parent material in the Bisley 3 watershed (400 m asl; Silver et al., 1999) and near the El Verde field station (500 m asl, Crow, 1980).

2.2. Experimental Design

We collected intact soil cores from two sites to examine how topographic controls on soil O_2 influence N gas emissions. First, we measured how N emissions varied across a macrotopographic gradient by collecting soils



from ridge, slope and valley positions (4 replicates per topographic position) from sites near the El Verde field station (Figure 1b) that are described in O'Connell et al. (2018). Macrotopography intact soil cores were collected and analyzed for N gas emissions at 0%, 10%, and 20% headspace O_2 in April of 2015. Second, we measured how N gas emissions varied across a microtopographic gradient, which consisted of 16 plots (1 m²) that differed in soil moisture and soil O_2 dynamics, as described by Hall et al. (2013). Because we did not measure soil O_2 availability at the time of this soil collection, the microtopography sites were binned by historic soil O_2 status into three categories: low, medium, and high soil O_2 based on Hall et al. (2013) (see Section 2.5 below). Microtopography intact soil cores were collected from Bisley watershed 3 (Figure 1c) and were analyzed for N gas emissions at 0%, 5%, 10%, and 20% headspace O_2 in January of 2015.

2.3. Soil Sampling

Mineral soils (0–10 cm) were collected using a split core sampler (3.4 cm width) and hammer auger in order to avoid compaction and to keep soil core structure intact. Intact soil cores were collected in triplicate (one for gas incubation measurements, one for soil incubation measurements, and one extra). Soils were collected in plastic sleeves, stored at room temperature in Ziploc bags and transported to the Cary Institute of Ecosystem Studies in Millbrook, NY USA the same day they were sampled. All laboratory analyses occurred within 3 days of soil sampling. Soil chemical analyses and N_2 , N_2O , and CO_2 emissions were measured at the Cary Institute, while soil O_2 and soil moisture data were collected in the field (see Section 2.5 below).

2.4. N₂, N₂O, and CO₂ Emissions

The Nitrogen Free Air Recirculation Method (N-FARM) laboratory-based gas flow incubation system was used to directly measure N_2 , N_2O , and CO_2 emissions from intact soil cores (Burgin et al., 2010). Soil cores were incubated at multiple headspace O_2 concentrations, and each incubation took place on a separate day using fresh soil cores in order to avoid soil nitrate depletion. Intact soil cores were removed from Ziploc bags prior to incubation and carefully placed upright into incubation jars to maintain the soil structure (Figure S1 in Supporting Information S1). Soil cores were sealed in jars and cyclically flushed and evacuated with a given helium-oxygen mixture (0%, 5%, 10%, or 20% headspace O_2) for ~16 hr to completely replace the N_2 atmosphere. After this flush, headspace samples were directly transferred via sample lines to Shimadzu GC 14 gas chromatographs equipped with thermal conductivity (for quantification of N_2 and CO_2) and electron capture (for N_2O) detectors (Burgin et al., 2010). To calculate an emissions flux, headspace samples were collected 0, 3, and 6 hr after headspace flushing concluded. Gas incubations were conducted on field moist soils (±1%). As a quality control measure, empty jars and autoclaved soil cores were run to ensure that the system was leak-free and that N_2 was not leaking from soil pores that were not adequately flushed prior to measurement. The method detection limit for the system was 3.98 mg $N_2 L^{-1}$, 0.0678 mg $N_2 O L^{-1}$, and 10.5 mg $CO_2 L^{-1}$, based on the standard deviation of six replicate standards (MDL = $T_{(af,0.01)} \times$ SD; Morse et al., 2015).

2.5. Soil Oxygen and Soil Moisture

We relied on previous year-long continuous soil O_2 measurements at the microtopography plots (Figure S2 in Supporting Information S1; Hall et al., 2013) to classify these sites into three categories: low, medium, and high soil O_2 . At the macrotopography plots, we measured soil moisture and soil O_2 for ~1 year using time-domain reflectometry (Campbell Scientific, Logan, Utah) and galvanic cell sensors (Apogee Instruments, Logan, Utah) housed in polyvinyl-chloride tubes installed at a depth of 10 cm. Soil O_2 levels and precipitation during the time of soil O_2 data collection (2010) and soil core collection (2015) were typical of this forest under non-drought conditions (Figure S2 in Supporting Information S1; O'Connell et al., 2018; Silver et al., 1999, 2013). In situ soil O_2 concentrations measured in the field are henceforth referred to as "soil O_2 " in order to differentiate them from incubation headspace O_2 manipulations, which are henceforth referred to as "headspace O_2 ." Gravimetric soil moisture was also determined on a subsample of all soil incubation cores at the time of sampling by calculating mass loss after oven drying soils at 105°C for 48 hr. Soil cores were stored in Ziploc bags at the time of collection until the time of analysis to maintain field moisture.

In situ soil O_2 data collected across the microtopography gradient several years prior to N measurements were used to bin the microtopography plots into groups that were comparable with headspace manipulated O_2 levels.



Plots with mean in situ soil O_2 concentrations from 5% to 10% are referred to as "Low," 10%–15% are referred to as "Medium," and 15%–20% are referred to as "High." Since no plot had a mean soil O_2 concentration <5%, a fourth O_2 category was omitted from our analyses (Figure 3). Given that soil O_2 patterns across the landscape remain relatively consistent over time, we assumed that soil O_2 emissions measured by Hall et al. (2013) were similar to those at the time of soil collection.

2.6. Statistical Analysis

Nitrogen gas emission data were not normally distributed, therefore we assessed differences between macroand micro-topographic positions using a non-parametric Wilcoxon Ranked sum test and examined relationships between variables using Spearman's Rho correlation coefficients. Where N emissions were undetectable, a value of 0.0001 was used to calculate $N_2:N_2O$ ratios for individual samples (Figure 4). Data are reported as means plus or minus the standard error.

3. Results

Dinitrogen dominated the N gas fluxes across all conditions (mean $N_2:N_2O = 275$). On average, N_2 emissions from all of our plots were 4.0 \pm 0.7 µg N g⁻¹ d⁻¹, compared to N₂O emissions of 0.02 \pm 0.003 µg N g⁻¹ d⁻¹. While laboratory incubation headspace O_2 concentrations had no significant effect on N_2 emissions, in situ soil O_2 concentrations, mediated by differences in macro- and micro-topographic position, had a significant influence on N_2 dynamics and variation in the $N_2:N_2O$ ratio (see Tables S1 and S2 in Supporting Information S1).

Emissions of N₂ were much greater than N₂O emissions in incubated cores sampled from macrotopography plots (mean N₂:N₂O = 70). Dinitrogen and N₂O emissions were higher from valleys compared to ridges and slopes (p < 0.05) at all headspace O₂ concentrations, with the exception of N₂O at 0% headspace O₂, which showed no difference between topographic positions (p = 0.6; Figure 2). Emissions of N₂O at 0% headspace O₂ were all zero, with the exception of a single outlier (0.14 µg N g⁻¹ d⁻¹). Although N₂ emission rates were highest in valleys, the N₂:N₂O was highest on aerated ridges, in part because N₂O emission rates were especially low there (Table 1). Soil moisture, which is related to in situ soil O₂ availability, was significantly higher in valleys (p < 0.0001; Table 1).

As with the macrotopography plots, N₂ emissions were much greater than N₂O emissions in incubated cores sampled from microtopography plots (mean N₂:N₂O = 410). The influence of microtopography on N₂ and N₂O emissions was more nuanced than that of macrotopography, with more consistent effects on N₂ than N₂O emissions (Figure 3). Microtopography plots were categorized by their mean in situ soil O₂ content, and plots categorized as having low mean soil O₂ had marginally higher soil moisture (p = 0.1; Table 2). Dinitrogen emissions were marginally higher (p < 0.1) from plots with low in situ soil O₂ concentrations at all incubation headspace O₂ concentrations, with the exception of the 5% headspace O₂ treatment (Figure 3). Microtopography had little effect on N₂O emissions. The only marginally significant difference in N₂O emissions ($0.02 \pm 0.01 \ \mu g \ M g^{-1} d^{-1}$, p = 0.06) compared to those with high or medium soil O₂ ($0.007 \pm 0.002 \ and 0.002 \pm 0.001 \ \mu g \ N g^{-1} d^{-1}$, respectively). The N₂:N₂O ratio was high from microtopography plots with low mean in situ soil O₂ (Table 2); where N₂ ($12.1 \pm 3.06 \ \mu g \ N g^{-1} d^{-1}$) was nearly three orders of magnitude higher than N₂O ($0.02 \pm 0.004 \ \mu g \ N g^{-1} d^{-1}$).

Soils were incubated under multiple headspace O_2 concentrations, which affected N_2O but not N_2 emissions. Soil collected from macrotopography sites had significantly higher N_2O emissions when incubated at 20% headspace O_2 (p = 0.01), compared to 0% and 10% headspace O_2 . Nitrous oxide emissions from microtopography sites were significantly higher when incubated at 5%, 10%, and 20% headspace O_2 (p < 0.0001) than when incubated at 5%, 10%, and 20% headspace O_2 (p < 0.0001) than when incubated with 0% headspace O_2 . The N_2 :N₂O ratio was significantly higher from 0% headspace O_2 incubations at both the micro- and macro-topography plots (p < 0.0001 and p = 0.009, respectively), as little N₂O was emitted during those incubations (Figure 4).

4. Discussion

4.1. The N₂ Emissions and the N₂:N₂O Ratio

We found that N_2 emissions dominated the N gas fluxes across all conditions. Mean N_2 emissions were three orders of magnitude higher than N_2O emissions and were highly variable across the landscape at both macro and

698961





Figure 2. Dinitrogen and nitrous oxide emissions from ridge, slope, and valley positions (macrotopography) near the El Verde Field Station measured at 0%, 10%, and 20% headspace O_2 .

micro scales. The magnitude of the N_2 flux make the inclusion of this measurement critical for accurate estimates of N budgets in lowland tropical forest. The three greatest uncertainties in most N budgets are N fixation, N storage in soil, and denitrification (Galloway et al., 2004); without accurate estimates of N_2 losses it is difficult to constrain inputs and storage. Furthermore, denitrification's variability speaks to the inadequacy of mass balance or static ratio approaches for estimating the N_2 emissions. Since methodological difficulties often preclude N_2 measurements in N budget studies, we show that soil O_2 concentrations co-vary with N_2 emissions, making this a useful parameter for estimating N_2 hot spots across the landscape, despite inadequate field measurement techniques.

The complex controls on N_2O emissions make it difficult to discern landscape-scale patterns and controls on the $N_2:N_2O$ ratio. A more nuanced understanding of controls on the $N_2:N_2O$ ratio can help inform soil management to reduce fluxes of N_2O to the atmosphere. Even while variable, the $N_2:N_2O$ ratio was dominated by a high numerator. Trends in this ratio were driven by N_2 in incubated cores sampled from microtopography plots, since N_2O showed little variability while N_2 varied greatly between plots (Table 2). However, in incubated cores sampled from macrotopography plots we saw opposing trends between $N_2:N_2O$ and N_2 , in part because N_2O production decreased more dramatically than N_2 on ridges (Table 1). In situ soil O_2 exerted stronger control on N_2 than on N_2O , the latter of which can be driven primarily by N availability particularly when produced by nitrification (Bremner & Blackmer, 1978; Firestone & Davidson, 1989) or affected by disturbances such as root mortality (Keller et al., 2000). While we report differences observed for field moist soils, soil saturation and subsequent entrapment of gases can affect observed emissions in gas incubations, particularly for N_2O , which is more likely to be reduced to N_2 when entrapped and thus underestimated (Letey et al., 1980). Broad scale patterns and controls of $N_2:N_2O$ will likely remain difficult to quantify due to these differential controls on the two gases and methodological limitations. These findings suggest scale-dependent controls on the $N_2:N_2O$ ratio, which provide





Figure 3. Dinitrogen and nitrous oxide emissions from slope plots that differed in their in situ mean soil O₂ (categorized as 5%, 10%, or 20%; microtopography) that were measured at 0%, 5%, 10% and 20% headspace O₂ in Bisley Watershed 3.

a framework for upscaling from variation within a given topographic position to the landscape scale (see section on Section 4.6).

4.2. Macrotopography and N Gas Emissions

Macrotopography was a very important control on N gas emissions. Both N₂O and N₂ emissions were highest in the valley samples regardless of incubation headspace O₂ concentration. Valleys are typically zones of high moisture, high soil O₂ consumption rates leading to low redox conditions (Liptzin et al., 2011; Silver et al., 1999), and can simultaneously have relatively high C and N concentrations that stimulate denitrification (McClain et al., 2003). While low points in the landscape often have finer soil texture than upland soils, valleys in the Bisley 3 and El Verde watersheds were found to be lower in both labile C and clay content than slopes and ridges (Hall et al., 2015; Hook & Burke, 2000; O'Connell et al., 2018; Pachepsky et al., 2001), thus it is unlikely that soil texture or carbon availability explain high N gas fluxes in valleys. Valleys at this site have been found to have higher soil moisture (O'Connell et al., 2018), which limits O₂ diffusion through the soil profile, than upland soils, suggesting that soil moisture and subsequent soil O₂ availability may be a key driving factor. Although valleys make up a small portion of the landscape in the forest, they are important hot spots for N gas production.

While valleys are important hot spots for N gas production, ridges can still produce N gases via anaerobic pathways in soil microsites. Although valleys produced the largest N₂ fluxes, the N₂:N₂O ratio was highest on aerated ridges. While ridges are indeed "aerated" at the bulk soil scale, Hall and Silver (2015) found that they actually had a greater mean Fe(II) content at the landscape scale (even relative to valleys), indicative of anaerobic microsites that could be driving N_2 production on ridges. This is consistent with the observation that ridges in this forest have been found to have greater aboveground biomass, root biomass, soil organic matter, and decomposition 21698961, 2023. I, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2022/G007210 by Jowa State University Library, Wiley Online Library on [24/01/023]. See the Terms

and Condi



Journal of Geophysical Research: Biogeosciences

10.1029/2022JG007210



Figure 4. Mean dinitrogen and nitrous oxide emissions from (a) macrotopography plots measured at 5%, 10%, or 20% headspace O_2 , and (b) microtopography plots measured at 0%, 5%, 10%, and 20% headspace O_2 .

activity (as indicated by extracellular enzymes) compared to other topographic positions, biological activity that would result in O_2 consumption in soil micropores (Hall et al., 2015; Scatena et al., 1993; Silver et al., 1994). The importance of these hot spots to ecosystem and landscape fluxes will depend on their areal extent and the magnitude of the denitrification rates that they support.

4.3. Microtopography and N Gas Emissions

The influence of microtopography (defined here as differences in mean soil O_2 availability among nearby plots) on N_2 and N_2O emissions was more nuanced than that of macrotopography. The microtopography effect was



Topographic position	$N_2 (ug N g^{-1} d^{-1})$	$N_2O (ug N g^{-1} d^{-1})$	N ₂ :N ₂ O	Mean in situ soil O ₂ (%)	Gravimetric soil moisture (%)
Ridge	0.12 (0.21)	0.0005 (0.0008)	240	18.47	41
Slope	0.24 (0.37)	0.003 (0.005)	80	16.07	43
Valley	2.9 (2.5)	0.04 (0.02)	73	0.04	50

Table 2

Overall Mean N_2 and N_2O Emissions (With Standard Deviation in Parentheses) and $N_2:N_2O$ Measured in Incubations at 0%, 5%, 10%, and 20% Headspace O_2 , and Mean In Situ Soil Oxygen and Gravimetric Soil Moisture for Microtopography Plots in Bisley Watershed 3 That Were Categorized Based on Their Mean In Situ Soil Oxygen Class (i.e., 5%, 10%, or 20%—No Plot had Mean Soil O_2 Below 5%)

Oxygen class	$N_2 (ug N g^{-1} d^{-1})$	$N_2O (ug N g^{-1} d^{-1})$	N ₂ :N ₂ O	Mean in situ soil O ₂ (%)	Gravimetric soil moisture (%)
High	4.34 (5.78)	0.015 (0.028)	289	18.17	45
Medium	3.74 (5.48)	0.012 (0.017)	312	12.36	44
Low	12.1 (10.6)	0.015 (0.015)	808	7.16	47

smaller than the macrotopography effect for both N_2 and N_2O emissions, suggesting that other factors such as C and mineral N availability may be important drivers here. Still, despite variability in N gas emissions and tropical forest heterogeneity (Townsend et al., 2008), we observed a relationship between long-term mean soil O_2 concentrations and one-time measurements of N gas emissions among soils that were otherwise very similar. While other factors are likely also important, these findings suggest that soil O_2 can serve as a useful integrating variable for predicting N fluxes at fine spatial scales (meters).

Dinitrogen emissions were greater from microtopography plots with low soil O_2 , and while N_2O emissions were also greater from plots with low soil O_2 when incubated at 20% headspace O_2 , N_2O emissions were more variable across soils when incubated under different headspace O_2 concentrations (Figure 3). These data suggest that soil O_2 is a primary driver of N_2 emissions, but that N_2O may have other drivers that dictate production. In other

tropical soils, N_2O emissions were correlated with nitrification rates (Matson & Vitousek, 1987), a process whose drivers may differ from that of denitrification. The decoupling of N_2O production from O_2 that we observed implicates roles for both nitrification and denitrification in N_2O production from this system.

4.4. Headspace Oxygen Availability and N Gas Emissions

While N_2 production was observed at all incubation headspace O_2 concentrations, reducing headspace O_2 to 0% stopped the production of N_2O completely at both types of sites (with the exception of a single outlier; Figure 4). This suggests that either N_2O from these sites is almost entirely produced from an aerobic pathway (i.e., nitrification) or that N_2O was produced by any number of pathways but was completely reduced to N_2 in an anaerobic environment. While N₂O emissions in riparian wetland soils from the temperate zone were found to be produced via aerobic pathways (Burgin & Groffman, 2012), those of temperate upland soils were found to decrease with increasing headspace O₂ (Kulkarni et al., 2014; Morse et al., 2015), implying that anaerobic processes were responsible for N₂O production. In agricultural, natural ecosystem, and culture studies, aerobic processes are often considered to be a primary source of N₂O, whereas denitrification is responsible only when soil moisture is high enough to create anaerobic microsites (Anderson & Levine, 1986; Bremner & Blackmer, 1978; Wrage et al., 2001). In riparian wetland and wet tropical forest soils, we expect the presence of such anaerobic microsites where N_2O is likely to be reduced to N_2 . These complexities highlight the need for more comparisons between temperate upland soils, where most denitrification studies have taken place, and wet environments, such as riparian wetlands and wet tropical forests, where N gas production likely differs as a result of high moisture and differences in soil aggregation/structure. Labile C and nitrate substrate are other potentially important mediating factors on N gas emissions (Weier et al., 1993); however, we did not explore those factors in this study.

The fact that N_2O , but not N_2 emissions, responded to changes in incubation headspace O_2 concentrations may reflect the location where production processes occur in the soil environment. Previous studies have found that intact soil core gas flow incubation system methods may overestimate N_2O emissions relative to N_2 (Morse et al., 2015), likely as a result of increased surface to volume ratios in the cores that reduce the residence time and potential for N_2O reduction. If N_2O production is occurring in macropores, which are exposed to O_2 during the soil core incubation, this can increase N_2O production. In contrast, if N_2 production is occurring in anaerobic microsites, this process is likely less vulnerable to short term changes in headspace O_2 caused by the core treatments and would thus be less responsive to induced headspace O_2 differences, as was observed in this study. This is consistent with our observation that N_2 emissions were sensitive to in situ O_2 concentrations (or topography) but not incubation headspace O_2 concentrations.

4.5. Tropical N Gas Emissions

Dinitrogen emissions were much greater than N_2O emissions (mean $N_2:N_2O = 275$). High N_2 emissions suggest high denitrification fluxes from this forest, likely resulting from soil aggregation of clays that allow for anaerobic microsites, moisture frequency in or above field capacity, high N mineralization rates, and some of the highest



decomposition rates on Earth (Cusack et al., 2010; Matson & Vitousek, 1987; Parton et al., 2007). Coupled nitrification-denitrification might explain why sites at Luquillo have been found to have high rates of net nitrification, yet available N pools dominated by NH_4^+ . Alternatively, this occurrence could be explained by dissimilatory nitrate reduction to ammonium (Silver et al., 2001). Feammox is another process that might lead to high N_2 emissions, as this process has been documented to emit N_2 at a rate similar to denitrification in this forest (Yang et al., 2012).

We report higher N_2 fluxes than have been detected using other methods in this forest. Our median and mean N_2 emissions were 0.14 and 1.09 µg N g⁻¹ d⁻¹, respectively, but ranged from 0 to 8.8 µg N g⁻¹ d⁻¹. Other estimates from this forest have been considerably lower. Using ¹⁵N tracer technique, Yang et al. (2014) estimated N_2 emission rates of 0.02 µg N g⁻¹ d⁻¹, while Silver et al. (2001) reported that they were unable to detect N_2 fluxes from ridge soils. Estimates by Yang et al. (2014) are similar to those we measured from ridges, but much lower than those we measured from valleys, consistent with our result that topography influences N_2 emissions. Our highest rates were from valleys and at low soil O_2 plots, where low redox potential and evidence of elevated N substrate availability (Scatena et al., 1993) likely created optimum conditions to produce N_2 . Differences in flux magnitude may result from either spatial/temporal heterogeneity or improved detection of N_2 using the NFARM.

Few measurements of denitrification have been made in other tropical forests and results vary widely. Kachenchart et al. (2012) found similar rates of N_2O (0.01 µg N g⁻¹ d⁻¹) from forest soils in Thailand, but lower rates of N_2 (0.05 µg N g⁻¹ d⁻¹) than we found in the LEF. Parsons et al. (1993) made measurements in Costa Rican soils and found N_2O (0.1 µg N g⁻¹ d⁻¹) emission rates that were similar in magnitude to N_2 emissions (0.09 µg N g⁻¹ d⁻¹). Cheng et al. (2014) and Zhu et al. (2013) found high rates of denitrification from subtropical forest soils in China (12.2 µg N g⁻¹ d⁻¹) and 5.38 µg N g⁻¹ d⁻¹, respectively), however, the former found N_2O dominated the N emissions pathway, whereas the latter found that N_2 was the dominant product. All measurements reported here used the acetylene inhibition method, which might explain the low N_2 estimates, due to inhibition of coupled nitrification-denitrification and incomplete diffusion of acetylene throughout the soil core (Groffman et al., 2006). More recently, Soper et al. (2018) reported low N gas emissions from a lowland tropical forest in Costa Rica using a mass balance approach. There is a clear need for more measurements of total N gas emissions from tropical soils, using improved methods that directly quantify N_2 emissions.

4.6. Scaling Up N Gas Emissions

There are two main challenges to producing ecosystem level N fluxes: (a) getting good estimates of instantaneous fluxes and (b) scaling up over space and time. The NFARM approach allowed us to overcome the first challenge, but our instantaneous fluxes range over several orders of magnitude depending on topographic position and sampling date (Tables 1 and 2). Thus, ecosystem N flux estimates are widely variable depending on the scaling approach taken.

Here we present topographically scaled values of N₂ production that are plausible from the perspective of mass balance and in the context of previous measurements. If we estimate the annual N_2 flux based on the mean of emissions measured near the El Verde Field Station, we get a flux in the hundreds of kg N ha⁻¹ yr⁻¹, which is obviously too high given that N inputs to this ecosystem are in the tens of kg N ha⁻¹ yr⁻¹ (Chestnut et al., 1999; Cusack et al., 2009). Since we found macrotopography to be a good indicator of denitrification, we used a spatially weighted mean based on the spatial extent of topographic positions across the landscape (i.e., 85% slope, 10% ridges, and only 5% valleys; Yang Lin, personal communication) and generate an annual N₂ flux of 93 kg N ha⁻¹ yr⁻¹. While accounting for topography cuts the annual flux estimate by more than half of that using the mean, it is likely still too high as average N fixation rates in tropical forests are 15-36 kg N ha⁻¹ yr⁻¹ (Cleveland et al., 1999). Since N_2 production rates are not constant over time, a better way to generate a flux might be to couple the topographic approach with soil O₂ dynamics, based on findings from headspace manipulation experiments. If we assume a threshold for N_2 production of <10% headspace O_2 (N_2 was significantly greater from plots with <10% in situ soil O₂; Figure 3), and that soil O₂ falls beneath this threshold 95% of the time in valleys, 5% on slopes and 0% on ridges (O'Connell et al., 2018), then we generate an N_2 flux of 33 kg N ha⁻¹ yr⁻¹. If we use our overall median rather than mean N₂ flux, our estimate based on soil O₂ dynamics increases to 32 kg N ha⁻¹ yr⁻¹ (see Table S3 in Supporting Information S1).

The large N_2 flux estimates we calculated are consistent with those of other tropical forests but have yet to be identified in the LEF. A recent study found that humid tropical forest soils average 17 kg N ha⁻¹ yr⁻¹ via



denitrification (Brookshire et al., 2017), while others have estimated 6–30 kg N ha⁻¹ yr⁻¹ (Bai et al., 2012; Houlton et al., 2006). These estimates are on par with measurements made from subtropical forests in China using a dual isotope technique, where they found denitrification rates of 2.4–21.7 kg N ha⁻¹ yr⁻¹ (Yu et al., 2019). Chestnut et al. (1999) estimated an N₂ flux of 1.3–3.75 kg N ha⁻¹ yr⁻¹ for the LEF, however, these measurements were made on sandier quartz diorite derived soils which have been found to display higher N₂O emission rates than that of clayey volcaniclastic soils (Bowden et al., 1992). Furthermore, Chestnut et al. (1999) calculated total N emissions by applying an N₂:N₂O (1:1) ratio to flux chamber measurements of N₂O, which based on variability in the N₂:N₂O ratio, is likely to generate estimates with a high degree of error.

For N₂O, if we use the overall mean flux, we generate an annual flux that is an order of magnitude higher than our spatially and temporally weighted estimates. When we use a spatially and temporally weighted mean, we estimate an N₂O flux of 0.45 kg N ha⁻¹ yr⁻¹. If we use the overall median flux, we generate a N₂O flux of 0.18 kg N ha⁻¹ yr⁻¹ (see Table S3 in Supporting Information S1). In comparison, Chestnut et al. (1999) estimated and N₂O flux of 0.6–1.7 kg N ha⁻¹ yr⁻¹ and Erickson et al. (2001) estimated an N oxide (NO and N₂O) flux ranging from 0.9 to 9 kg N ha⁻¹ yr⁻¹ in this forest.

Together these estimates highlight the importance of quantifying N_2 emissions in order to adequately estimate the N flux from tropical forest soils. These findings emphasize how the proportion of these gases emitted from soils varies across both space and time. Despite variability in N gas fluxes and heterogeneity across tropical landscapes (Townsend et al., 2008), we show that field measurements across naturally occurring ecosystem gradients can help to build a more robust understanding of tropical N cycling.

5. Conclusions

These are the first unamended direct measurements of N_2 fluxes from tropical forest soils. We report some of the highest N_2 flux rates for tropical forests and found that in situ soil O_2 dynamics have a strong influence on N_2 emissions, which dominate the N gas loss pathway. Soil O_2 dynamics have a greater effect on N gas emissions at the macro-than at the micro-scale. Valleys appear to be a hot spot for both N_2O and N_2 production, while smaller scale microtopographic variation in soil O_2 across the rest of the landscape influences N_2 fluxes much more than N_2O . Incubation headspace O_2 influenced N_2O but not N_2 production along both macro and micro-topographic gradients, suggesting a potential difference in either the physical location in which each gas is being produced or the process that is producing them. It was difficult to discern controls on the N_2 . Oratio, likely due to differential controls on the two gases. Dinitrogen emissions appear to be driven by characteristic differences in soil O_2 dynamics, whereas N_2O emissions may have alternate or more complex drivers. These results suggest that it is not possible to model the emissions of one gas based on measurements of the other (Chestnut et al., 1999), which complicates our ability to assess and model N transformations across tropical forest landscapes. More sophisticated modeling efforts should be employed to accurately capture the effect of hot spots and hot moments on annual N gas emission fluxes. Furthermore, the distribution and intensity of these hot spots are likely sensitive to multiple components of global environmental change.

These findings confirm the theoretical conclusion that N gas emissions from tropical forests may in fact be large (Davidson et al., 2000; C. Wang et al., 2017) and support the idea that there is wide variation in the nature of N cycling and losses in lowlands tropical forests (Soper et al., 2018). The magnitude of emissions estimated here are higher than inputs, this finding may speak to a need for better estimates of tropical forest inputs, which currently rely mainly on a single analysis (Cleveland et al., 1999). Increasing direct measurement of N inputs and outputs from tropical forests are still necessary to understanding N budgets in these understudied ecosystems.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

Summarized data for this publication are available in Tables S1 and S2 of Supporting Information S1. Non-aggregated data can be made available by authors upon request or is publicly available on the Zenodo data repository (https://doi.org/10.5281/zenodo.7411198).



Acknowledgments

We want to thank Geoffrey Schwaner, Josh Brown, Ryan Salladay, Ruby Ho, Wade Simmons, Michelle Wong, and Lisa Martell for their help in the laboratory and the field. Maya Almaraz and Stephen Porder received funding from the NIH Initiative for Maximizing Student Development (R25GM083270), and the NSF Doctoral Dissertation Improvement Grant (GR5260021). Whendee Silver, Yang Lin and Christine O'Connell were supported by the NSF Luquillo LTER (DEB-1546686) to the University of Puerto Rico, and DOE-TES-DE-FOA-0000749. All authors were supported by the NSF Luquillo Critical Zone Observatory (EAR-1331841) to the University of New Hampshire.

References

- Almaraz, M., Groffman, P. M., & Porder, S. (2019). Effects of changes in nitrogen availability on nitrogen gas emissions in a tropical forest during a drought. Journal of Geophysical Research: Biogeosciences, 124(9), 2917–2926. https://doi.org/10.1029/2018jg004851
- Almaraz, M., Wong, M. Y., & Yang, W. H. (2020). Looking back to look ahead: A vision for soil denitrification research. Ecology, 101(1), e02917. https://doi.org/10.1002/ecy.2917
- Anderson, I. C., & Levine, J. S. (1986). Relative rates of nitric oxide and nitrous oxide production by nitrifiers, denitrifiers, and nitrate respirers. *Applied and Environmental Microbiology*, 51(5), 938–945. https://doi.org/10.1128/aem.51.5.938-945.1986
- Bai, E., Houlton, B. Z., & Wang, Y. P. (2012). Isotopic identification of nitrogen hotspots across natural terrestrial ecosystems. *Biogeosciences*, 9(8), 3287–3304. https://doi.org/10.5194/bg-9-3287-2012
- Bowden, W. B., McDowell, W. H., Asbury, C. E., & Finley, A. M. (1992). Riparian nitrogen dynamics in two geomorphologically distinct tropical rain forest watersheds: Nitrous oxide fluxes. *Biogeochemistry*, 18(2), 77–99. https://doi.org/10.1007/bf00002704
- Bremner, J. M., & Blackmer, A. M. (1978). Nitrous oxide: Emission from soils during nitrification of fertilizer nitrogen. *Science*, 199(4326), 295–296. https://doi.org/10.1126/science.199.4326.295
- Bremner, J. M., & Blackmer, A. M. (1981). Terrestrial nitrification as a source of atmospheric nitrous oxide [Soil, aquatic system].
- Brookshire, E. N. J., Gerber, S., Greene, W., Jones, R. T., & Thomas, S. A. (2017). Global bounds on nitrogen gas emissions from humid tropical forests. *Geophysical Research Letters*, 44(5), 2502–2510. https://doi.org/10.1002/2017gl072867
- Burgin, A. J., Groffman, P., & Lewis, D. (2010). Factors regulating denitrification in a riparian wetland. Soil Science Society of America Journal, 74(5), 1826–1833. https://doi.org/10.2136/sssaj2009.0463
- Burgin, A. J., & Groffman, P. M. (2012). Soil O₂ controls denitrification rates and N₂O yield in a riparian wetland. *Journal of Geophysical Research*, 117(G1). https://doi.org/10.1029/2011jg001799
- Butterbach-Bahl, K., Baggs, E. M., Dannenmann, M., Kiese, R., & Zechmeister-Boltenstern, S. (2013). Nitrous oxide emissions from soils: How well do we understand the processes and their controls? *Philosophical Transactions of the Royal Society B: Biological Sciences*, 368(1621), 20130122. https://doi.org/10.1098/rstb.2013.0122
- Cheng, Y., Wang, J., Liu, Y., & Cai, Z. C. (2014). Litter decomposition reduces either N2O or NO production in strongly acidic coniferous and broad-leaved forest soils under anaerobic conditions. *Journal of Soils and Sediments*, 14(3), 549–557. https://doi.org/10.1007/s11368-013-0795-y
- Chestnut, T. J., Zarin, D. J., McDowell, W. H., & Keller, M. (1999). A nitrogen budget for late-successional hillslope tabonuco forest, Puerto Rico. Biogeochemistry, 46(1–3), 85–108. https://doi.org/10.1007/bf01007575
- Cleveland, C. C., Townsend, A. R., Schimel, D. S., Fisher, H., Howarth, R. W., Hedin, L. O., et al. (1999). Global patterns of terrestrial biological nitrogen (N₂) fixation in natural ecosystems. *Global Biogeochemical Cycles*, 13(2), 623–645. https://doi.org/10.1029/1999gb900014
- Crow, T. R. (1980). A rainforest chronicle: A 30-year record of change in structure and composition at El Verde, Puerto Rico (pp. 42–55). Biotropica.
- Cusack, D. F., Silver, W., & McDowell, W. H. (2009). Biological nitrogen fixation in two tropical forests: Ecosystem-level patterns and effects of nitrogen fertilization. *Ecosystems*, 12(8), 1299–1315. https://doi.org/10.1007/s10021-009-9290-0
- Cusack, D. F., Torn, M. S., McDowell, W. H., & Silver, W. L. (2010). The response of heterotrophic activity and carbon cycling to nitrogen additions and warming in two tropical soils. *Global Change Biology*, 16(9), 2555–2572. https://doi.org/10.1111/j.1365-2486.2009.02131.x
- Davidson, E. A., Keller, M., Erickson, H. E., Verchot, L. V., & Veldkamp, E. (2000). Testing a conceptual model of soil emissions of nitrous and nitric oxides: Using two functions based on soil nitrogen availability and soil water content, the hole-in-the-pipe model characterizes a large fraction of the observed variation of nitric oxide and nitrous oxide emissions from soils. *BioScience*, 50(8), 667–680. https://doi.org/10.1641/0006-3 568(2000)050[0667:tacmos]2.0.co;2
- Erickson, H., Keller, M., & Davidson, E. A. (2001). Nitrogen oxide fluxes and nitrogen cycling during postagricultural succession and forest fertilization in the humid tropics. *Ecosystems*, 4(1), 67–84. https://doi.org/10.1007/s100210000060
- Firestone, M. K., & Davidson, E. A. (1989). Microbiological basis of NO and N₂O production and consumption in soil. In M. O. Andreae & D. S. Schimel (Eds.), *Exchange of trace gases between terrestrial ecosystems and the atmosphere* (pp. 7–21). Wiley.
- Firestone, M. K., Firestone, R. B., & Tiedje, J. M. (1980). Nitrous oxide from soil denitrification: Factors controlling its biological production. Galloway, J. N., Dentener, F. J., Capone, D. G., Boyer, E. W., Howarth, R. W., Seitzinger, S. P., et al. (2004). Nitrogen cycles: Past, present, and future. *Biogeochemistry*, 70(2), 153–226. https://doi.org/10.1007/s10533-004-0370-0
- Garcia-Montino, A. R., Warner, G. S., Scatena, F. N., & Civco, D. L. (1996). Rainfall, runoff and elevation relationships in the Luquillo Mountains of Puerto Rico. Caribbean Journal of Science, 32, 413–424.
- Goreau, T. J., Kaplan, W. A., Wofsy, S. C., McElroy, M. B., Valois, F. W., & Watson, S. W. (1980). Production of NO₂-and N₂O by nitrifying bacteria at reduced concentrations of oxygen. *Applied and Environmental Microbiology*, 40(3), 526–532. https://doi.org/10.1128/aem.40.3.526-532.1980
- Groffman, P. M., Altabet, M. A., Böhlke, J. K., Butterbach-Bahl, K., David, M. B., Firestone, M. K., et al. (2006). Methods for measuring denitrification: Diverse approaches to a difficult problem. *Ecological Applications*, 16(6), 2091–2122. https://doi.org/10.1890/1051-0761(2006)016[2091:mfmdda]2.0.co;2
- Groffman, P. M., & Tiedje, J. M. (1989). Denitrification in north temperate forest soils: Relationships between denitrification and environmental factors at the landscape scale. Soil Biology and Biochemistry, 21(5), 621–626. https://doi.org/10.1016/0038-0717(89)90054-0
- Hall, S. J., McDowell, W. H., & Silver, W. L. (2013). When wet gets wetter: Decoupling of moisture, redox biogeochemistry, and greenhouse gas fluxes in a humid tropical forest soil. *Ecosystems*, *16*(4), 1–14. https://doi.org/10.1007/s10021-012-9631-2
- Hall, S. J., & Silver, W. L. (2015). Reducing conditions, reactive metals, and their interactions can explain spatial patterns of surface soil carbon in a humid tropical forest. *Biogeochemistry*, 125(2), 149–165. https://doi.org/10.1007/s10533-015-0120-5
- Hall, S. J., Silver, W. L., Timokhin, V. I., & Hammel, K. E. (2015). Lignin decomposition is sustained under fluctuating redox conditions in humid tropical forest soils. *Global Change Biology*, 21(7), 2818–2828. https://doi.org/10.1111/gcb.12908
- Harris, N. L., Lugo, A. E., Brown, S., & Heartsill-Scalley, T. (2012). Luquillo experimental forest: Research history and opportunities. Experimental Forest and Range EFR-1 (p. 152). USDA Forest Service.
- Hook, P. B., & Burke, I. C. (2000). Biogeochemistry in a shortgrass landscape: Control by topography, soil texture, and microclimate. *Ecology*, 81(10), 2686–2703. https://doi.org/10.1890/0012-9658(2000)081[2686:biaslc]2.0.co;2
- Houlton, B. Z., Sigman, D. M., & Hedin, L. O. (2006). Isotopic evidence for large gaseous nitrogen losses from tropical rainforests. Proceedings of the National Academy of Sciences, 103(23), 8745–8750. https://doi.org/10.1073/pnas.0510185103

- Kachenchart, B., Jones, D. L., Gajaseni, N., Edwards-Jones, G., & Limsakul, A. (2012). Seasonal nitrous oxide emissions from different land uses and their controlling factors in a tropical riparian ecosystem. Agriculture, Ecosystems and Environment, 158, 15–30. https://doi.org/10.101 6/j.agee.2012.05.008
- Keller, M., Weitz, A. M., Bryan, B., Rivera, M. M., & Silver, W. L. (2000). Soil-atmosphere nitrogen oxide fluxes: Effects of root disturbance. Journal of Geophysical Research, 105(D14), 17693–17698. https://doi.org/10.1029/2000jd900068
- Kulkarni, M. V., Burgin, A. J., Groffman, P. M., & Yavitt, J. B. (2014). Direct flux and ¹⁵N tracer methods for measuring denitrification in forest soils. *Biogeochemistry*, 117(2–3), 359–373. https://doi.org/10.1007/s10533-013-9876-7
- Letey, J., Jury, W. A., Hadas, A., & Valoras, N. (1980). Gas diffusion as a factor in laboratory incubation studies on denitrification. American Society of Agronomy, Crop Science Society of America, and Soil Science Society of America, 9(2), 223–227. https://doi.org/10.2134/ jeg1980.00472425000900020012x
- Liptzin, D., Silver, W. L., & Detto, M. (2011). Temporal dynamics in soil oxygen and greenhouse gases in two humid tropical forests. *Ecosystems*, 14(2), 171–182. https://doi.org/10.1007/s10021-010-9402-x
- Matson, P. A., & Vitousek, P. M. (1987). Cross-system comparisons of soil nitrogen transformations and nitrous oxide flux in tropical forest ecosystems. *Global Biogeochemical Cycles*, 1(2), 163–170. https://doi.org/10.1029/gb001i002p00163
- McClain, M. E., Boyer, E. W., Dent, C. L., Gergel, S. E., Grimm, N. B., Groffman, P. M., et al. (2003). Biogeochemical hot spots and hot moments at the interface of terrestrial and aquatic ecosystems. *Ecosystems*, 6(4), 301–312. https://doi.org/10.1007/s10021-003-0161-9
- Morse, J. L., Durán, J., & Groffman, P. M. (2015). Soil denitrification fluxes in a Northern Hardwood forest: The importance of snowmelt and implications for ecosystem N budgets. *Ecosystems*, 18(3), 520–532. https://doi.org/10.1007/s10021-015-9844-2
- O'Connell, C. S., Ruan, L., & Silver, W. L. (2018). Drought drives rapid shifts in tropical rainforest soil biogeochemistry and greenhouse gas emissions. *Nature Communications*, 9(1), 1348. https://doi.org/10.1038/s41467-018-03352-3
- Oren, R., Ellsworth, D. S., Johnsen, K. H., Phillips, N., Ewers, B. E., Maier, C., et al. (2001). Soil fertility limits carbon sequestration by forest ecosystems in a CO₂-enriched atmosphere. *Nature*, 411(6836), 469–472. https://doi.org/10.1038/35078064
- Pachepsky, Y. A., Timlin, D. J., & Rawls, W. J. (2001). Soil water retention as related to topographic variables. Soil Science Society of America Journal, 65(6), 1787–1795. https://doi.org/10.2136/sssaj2001.1787
- Parkin, T. B. (1987). Soil microsites as a source of denitrification variability. Soil Science Society of America Journal, 51(5), 1194–1199. https://doi.org/10.2136/sssaj1987.03615995005100050019x
- Parsons, W. F., Mitre, M. E., Keller, M., & Reiners, W. A. (1993). Nitrate limitation of N2O production and denitrification from tropical pasture and rain forest soils. *Biogeochemistry*, 22(3), 179–193. https://doi.org/10.1007/bf00000646
- Parton, W., Silver, W. L., Burke, I. C., Grassens, L., Harmon, M. E., Currie, W. S., et al. (2007). Global-scale similarities in nitrogen release patterns during long-term decomposition. *Science*, 315(5810), 361–364. https://doi.org/10.1126/science.1134853
- Pett-Ridge, J., Silver, W. L., & Firestone, M. K. (2006). Redox fluctuations frame microbial community impacts on N-cycling rates in a humid tropical forest soil. *Biogeochemistry*, 81(1), 95–110. https://doi.org/10.1007/s10533-006-9032-8
- Ponnamperuma, F. N. (1972). The chemistry of submerged soils. Advances in Agronomy, 24, 29-96.
- Poth, M., & Focht, D. D. (1985). ¹⁵N kinetic analysis of N₂O production by Nitrosomonas europaea: An examination of nitrifier denitrification. *Applied and Environmental Microbiology*, 49(5), 1134–1141. https://doi.org/10.1128/aem.49.5.1134-1141.1985
- Scatena, F. N. (1989). An introduction to the physiography and history of the Bisley experimental watersheds in the Luquillo Mountains of Puerto Rico, General Technical Report, SO-72 (p. 22). Southern Forest Experiment Station, U.S. Department of Agriculture.
- Scatena, F. N., Silver, W., Siccama, T., Johnson, A., & Sanchez, M. J. (1993). Biomass and nutrient content of the Bisley experimental watersheds, Luquillo Experimental Forest, Puerto Rico, before and after Hurricane Hugo, 1989. *Biotropica*, 25(1), 15–27. https://doi.org/10.2307/2388975
- Silver, W. L., Herman, D. J., & Firestone, M. K. (2001). Dissimilatory nitrate reduction to ammonium in upland tropical forest soils. *Ecology*, 82(9), 2410–2416. https://doi.org/10.1890/0012-9658(2001)082[2410:dnrtai]2.0.co;2
- Silver, W. L., Liptzin, D., & Almaraz, M. (2013). Soil redox dynamics and biogeochemistry along a tropical elevational gradient. In G. González, M. R. Willig, & R. B. Waide (Eds.), 2013: Ecological gradient analyses in a tropical landscape, Ecological Bulletin (Vol. 4, pp. 195–209).
- Silver, W. L., Lugo, A. E., & Keller, M. (1999). Soil oxygen availability and biogeochemistry along rainfall and topographic gradients in upland wet tropical forest soils. *Biogeochemistry*, 44(3), 301–328. https://doi.org/10.1007/bf00996995
- Silver, W. L., Scatena, F. N., Johnson, A. H., Siccama, T. G., & Sanchez, M. J. (1994). Nutrient availability in a montane wet tropical forest: Spatial patterns and methodological considerations. *Plant and Soil*, 164(1), 129–145. https://doi.org/10.1007/bf00010118
- Soper, F. M., Taylor, P. G., Wieder, W. R., Weintraub, S. R., Cleveland, C. C., Porder, S., & Townsend, A. R. (2018). Modest gaseous nitrogen losses point to conservative nitrogen cycling in a lowland tropical forest watershed. *Ecosystems*, 21(5), 901–912. https://doi.org/10.1007/ s10021-017-0193-1
- Stein, L. Y. (2019). Insights into the physiology of ammonia-oxidizing microorganisms. *Current Opinion in Chemical Biology*, 49, 9–15. https://doi.org/10.1016/j.cbpa.2018.09.003
- Templer, P. H., Silver, W. L., Pett-Ridge, J. M., DeAngelis, K., & Firestone, M. K. (2008). Plant and microbial controls on nitrogen retention and loss in a humid tropical forest. *Ecology*, 89(11), 3030–3040. https://doi.org/10.1890/07-1631.1
- Townsend, A. R., Asner, G. P., & Cleveland, C. C. (2008). The biogeochemical heterogeneity of tropical forests. *Trends in Ecology & Evolution*, 23(8), 424–431. https://doi.org/10.1016/j.tree.2008.04.009
- Vitousek, P. M., & Matson, P. A. (1988). Nitrogen transformations in a range of tropical forest soils. Soil Biology and Biochemistry, 20(3), 361–367. https://doi.org/10.1016/0038-0717(88)90017-x
- Vitousek, P. M., & Sanford, R. L. (1986). Nutrient cycling in moist tropical forest. Annual Review of Ecology and Systematics, 17(1), 137–167. https://doi.org/10.1146/annurey.es.17.110186.001033
- Wang, C., Houlton, B. Z., Dai, W., & Bai, E. (2017). Growth in the global N₂ sink attributed to N fertilizer inputs over 1860 to 2000. Science of the Total Environment, 574, 1044–1053. https://doi.org/10.1016/j.scitotenv.2016.09.160
- Wang, Y. P., & Houlton, B. Z. (2009). Nitrogen constraints on terrestrial carbon uptake: Implications for the global carbon-climate feedback. Geophysical Research Letters, 36(24), L24403. https://doi.org/10.1029/2009gl041009
- Weier, K. L., Doran, J. W., Power, J. F., & Walters, D. T. (1993). Denitrification and the dinitrogen/nitrous oxide ratio as affected by soil water, available carbon, and nitrate. Soil Science Society of America Journal, 57(1), 66–72. https://doi.org/10.2136/sssaj1993.03615995005700010013x
- Wrage, N., Velthof, G. L., Van Beusichem, M. L., & Oenema, O. (2001). Role of nitrifier denitrification in the production of nitrous oxide. Soil Biology and Biochemistry, 33(12), 1723–1732. https://doi.org/10.1016/s0038-0717(01)00096-7
- Yang, W. H., McDowell, A. C., Brooks, P. D., & Silver, W. L. (2014). New high precision approach for measuring ¹⁵N–N₂ gas fluxes from terrestrial ecosystems. *Soil Biology and Biochemistry*, 69, 234–241. https://doi.org/10.1016/j.soilbio.2013.11.009
- Yang, W. H., Weber, K. A., & Silver, W. L. (2012). Nitrogen loss from soil through anaerobic ammonium oxidation coupled to iron reduction. *Nature Geoscience*, 5(8), 538–541. https://doi.org/10.1038/ngeo1530



- Yu, L., Mulder, J., Zhu, J., Zhang, X., Wang, Z., & Dörsch, P. (2019). Denitrification as a major regional nitrogen sink in subtropical forest catchments: Evidence from multi-site dual nitrate isotopes. *Global Change Biology*, 25(5), 1765–1778. https://doi.org/10.1111/gcb.14596 Zhu, J., Mulder, J., Solheimslid, S. O., & Dörsch, P. (2013). Functional traits of denitrification in a subtropical forest catchment in China with high atmogenic N deposition. *Soil Biology and Biochemistry*, 57, 577–586. https://doi.org/10.1016/j.soilbio.2012.09.017
- Zhuang, Q., Lu, Y., & Chen, M. (2012). An inventory of global N₂O emissions from the soils of natural terrestrial ecosystems. *Atmospheric Environment*, 47, 66–75. https://doi.org/10.1016/j.atmosenv.2011.11.036